Author's response to reviews

Title: Analysis of geographic distribution of HFRS in Liaoning Province between 2000 and 2005

Authors:

Hualiang Lin (linhualiang2002@163.com)
Qiyong Liu (liuqiyong@icdc.cn)
Junqiao Guo (junqiao_quo@163.com)
Jibo Zhang (daoshou-19811111@163.com)
Jinfeng Wang (wangjf@lreis.ac.cn)
Huaxin Chen (chenhuaxin@icdc.cn)

Version: 6 Date: 22 March 2007

Author's response to reviews:

Dear editors,

In order to perfect our manuscript, we are always bettering it, and now we want to submit one new version, which has been revised as follows:

1. One sentence (the last sentence in Methods part of Abstract) may have some phraseological error, so we corrected it.

2. The last sentence of the Results of the Abstract was also collected for the expressing error.

3. In the Background, we changed environmental area, climatic area into environmental factors and climatic factors, respectively.

And there were also several small changes like some word spellings.

Thank you for your attention.

Best wishes,

Yours

Lin.